Author's response to reviews

Title: Physical Activity is Associated with Reduced Risk of Esophageal Cancer, Particularly Esophageal Adenocarcinoma: A Systematic Review and Meta-Analysis

Authors:

Siddharth Singh (singh.siddharth2@mayo.edu)
Swapna Devanna (Devanna.Swapna@mayo.edu)
Jithinraj Edakkanambeth Varayil (EdakkanambethVarayil.Jithinraj@mayo.edu)
Mohammad H. Murad (Murad.Mohammad@mayo.edu)
Prasad G. Iyer (Iyer.Prasad@mayo.edu)

Version: 2
Date: 15 January 2014

Author's response to reviews:

Changes made to manuscript:
- Added emails of all authors to title page
- Added Acknowledgement section
- Provided Ethics statement
- Added Figure Legends after References